Author's response to reviews

Title: Do Pediatric Emergency Departments Pose a Risk of Infection?

Authors:

Caroline Quach (caroline.quach@mcgill.ca)
Dorothy L Moore (dorothy.moore@muhc.mcgill.ca)
Francine M Ducharme (francine.m.ducharme@umontreal.ca)
Dominic Chalut (dominic.chalut@muhc.mcgill.ca)

Version: 3 Date: 6 January 2011

Author's response to reviews:

As requested, the following changes were made:
1. Affiliations were reformatted to meet the required standards
2. The abstract in the manuscript is exactly the same as the one in the submission system
3. References' format has been modified as requested
4. Highlighting/tracking was removed as requested
5. The visible vertical lines in the tables are actually not visible - they just indicate the separation between cells and do not show on the printed version. I am unable to make them look fainter in the word document
6. The merged cells have been removed and reformatted